Author's response to reviews

Title: Giant cutaneous horn in an African woman: a case report

Authors:

Peter M Nthumba (nthumba@gmail.com)

Version: 6 Date: 29 November 2007

Author's response to reviews:

Title Page: Peter M. Nthumba (was Peter Muli Nthumba)

Abstract: Entire Abstract restructured to incorporate `Introduction¿, `Case Presentation¿, and `Conclusions¿.

Conclusions: Added `and superstition¿

Consent placed after Authors¿ contributions.

References

1. Changed upper case (C H) `Cutaneous Horns¿ to `cutaneous horns¿
2. Replaced et al with additional authors: Requena L, Fanburg-Smith JC, Mukensnabl P, Hes O, Cada F. Also changed upper case `C H S W G S¿ to lower case `c h s w g s¿ in the title, `Gigantic cutaneous horns of the scalp. Lesions with a gross similarity to the horns of animals: A report of four cases.¿
3. Changed upper case `C C H¿ to lower case `c c h¿ in Cornu cutaneum at an unusual site.